Author’s response to reviews

Title: Prolonged persistence of IgM against dengue virus detected by commonly used commercial assays

Authors:
Guey Chuen Perng (gperng@mail.ncku.edu.tw)
Yu-Wen Chien (yuwenchien@mail.ncku.edu.tw)
Zi-Hu Liu (acosline@hotmail.com.tw)
Fan-Chen Tseng (franc.tseng@gmail.com)
Tzu-Chuan Ho (anywayhowtodo@gmail.com)
How-Ran Guo (hrguo@mail.ncku.edu.tw)
Nai-Ying Ko (nyko@mail.ncku.edu.tw)
Wen-Chien Ko (winston@mail.ncku.edu.tw)

Version: 3 Date: 22 Mar 2018

Author’s response to reviews:

The editor's suggestions have been added to text in the proper section accordingly. A final revised and clean copy has been submitted through the editorial manager system.